Author's response to reviews

Title: Factors influencing sustained implementation and fidelity thereof and related student outcomes of an evidence-based national HIV prevention program

Authors:

Bo Wang (bwang@med.wayne.edu)
Bonita Stanton (bstanton@med.wayne.edu)
Lynette Deveaux (lyndevexa@yahoocom)
Maxwell Poitier (m.g.poitier@gmail.com)
Sonja Lunn (slunn@coralwave.com)
Veronica Koci (vkoci@med.wayne.edu)
Richard Adderley (radderley@gmail.com)
Linda Kaljee (lkaljee@yahoo.com)
Sharon Marshall (smarshal@med.wayne.edu)
Xiaoming Li (xiaoli@med.wayne.edu)
Glenda Rolle (grolle54@gmail.com)

Version: 2
Date: 14 October 2014

Author's response to reviews:

Dr. Anne Sales
Editor-in-Chief, Implementation Science
University of Michigan School of Nursing
400 North Ingalls Building
Ann Arbor, MI 48109-5482
USA

Dr. Michel Wensing
Editor-in-Chief, Implementation Science
Institute for Quality in Healthcare
Radboud University Nijmegen Medical Centre,
Geert Grooteplein noord 21
Nijmegen 6525 EZ
Netherlands

Dear Dr. Sales and Dr. Wensing:

We thank you for the opportunity to make the changes outlined in your letter of October 13, 2014. Specifically, we have made the following changes. We now include:
- a title page as page 1. The title page includes the title, authors, author affiliations and corresponding author details.
- a Conclusions section as the last section of the text.
- a Competing Interests section at the end of the manuscript, stating correctly: "The authors declare that they have no competing interests."
- an Acknowledgement Section at the end of the manuscript before the reference list.
- an Authors’ Contributions section at the end of the manuscript, before the reference list.

We shall upload the figures as separate files via the online submission system.

Sincerely yours,

Corresponding author:
Bonita F. Stanton, M.D.
Vice Dean for Research
School of Medicine
Wayne State University
313-577-9553 (office)
313-577-9399 (FAX)
Email: bstanton@med.wayne.edu